Author's response to reviews

**Title:** The peroxisome proliferator-activated receptor (PPAR) alpha agonist fenofibrate maintains bone mass, while the PPAR gamma agonist pioglitazone exaggerates bone loss, in ovariectomized rats

**Authors:**

Astrid K Stunes (kamilla.stunes@ntnu.no)
Irene Westbroek (westbroekirene@gmail.com)
Björn I Gustafsson (bjorn.gustafsson@ntnu.no)
Reidar Fossmark (reidar.fossmark@ntnu.no)
Jan H Waarsing (e.waarsing@erasmusmc.nl)
Erik F Eriksen (e.f.eriksen@medisin.uio.no)
Christian Petzold (christiane.petzold@odont.uio.no)
Janne E Reseland (j.e.reseland@odont.uio.no)
Unni Syversen (unni.syversen@ntnu.no)

**Version:** 4 **Date:** 24 May 2011

**Author's response to reviews:**

Editor-in-Chief

Dated: May 24th, 2011 Our ref.: Astrid Kamilla Stunes

Dear Editor

Enclosed, please find our revised manuscript with the requested formatting changes.

Requested Formatting changes to be made:

-----------------------------------

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

This is now corrected

Abstract: Please include Abstract on manuscript.

This is now included in the manuscript

Change 'Introduction' to 'Background' and 'Material and Methods' to 'Methods.'

This is now changed
Structure: Please include 'Conclusion' after 'Discussion'
This is now added to the manuscript

Authors' contributions: Please indicate that all authors read and approved the final manuscript.
This is now confirmed in the manuscript

References: Please provide functioning link for http://www2.eur.nl/fgg/orthopedics/downloads
The link is changed to a functional one

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com
This is done and added to the title page

Line numbers: Please remove these.
These have been removed

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have carefully proofread the manuscript to avoid typographical errors

Address correspondence to:
Astrid Kamilla Stunes
Department of Cancer Research and Molecular Medicine
Faculty of Medicine, MTFS
Norwegian University of Science and Technology, NTNU
Olav Kyrres gt. 9
N-7489 Trondheim
Norway
Phone: +47 72825357
Fax: +47 73598801
Email: kamilla.stunes@ntnu.no

On behalf of the authors
Astrid Kamilla Stunes